Author's response to reviews

Title: Parijaat: Improving quality of childbirth services in public health facilities of Rajasthan, India

Authors:

Kirti Iyengar (kirtiyengar@gmail.com)
Sharad D. Iyengar (sdiyengar@gmail.com)
Motilal Jain (motilaljiain2001@yahoo.co.uk)
Sunil Thomas (jacob@unfpa.org)
Kalpana Dashora (arth.smp@gmail.com)
William Liu (will.c.liu@gmail.com)
Paramsukh Saini (pssaini.83@gmail.com)
Rajesh Dattatreya (arth.smp@gmail.com)
Indrani Parker (indrani.parker@gmail.com)

Version: 4 Date: 22 December 2013

Author’s response to reviews: see over
Author's response to reviews

Title: Parijaat: Improving quality of childbirth services in public health facilities of Rajasthan, India

Authors:

Kirti Iyengar (kirtiyengar@gmail.com)
Sharad D. Iyengar (sdiyengar@gmail.com)
Motilal Jain (motilaljain2001@yahoo.co.uk)
Sunil Thomas (jacob@unfpa.org)
Kalpana Dashora (arth.smp@gmail.com)
William Lu (will.c.liu@gmail.com)
Paramsukh Saini (pssaini.83@gmail.com)
Rajesh Dattatreya (arth.smp@gmail.com)
Indrani Parker (indrani.parker@gmail.com)

Version: 3 Date: 18 December 2013

Author's response to reviews: see over
December 2013

Members of Editorial Board,
BM C Pregnancy and Childbirth

Dear Editorial Team,

We are submitting a revision of the manuscript titled, "Parijaat: Improving quality of childbirth services in public health facilities of Rajasthan, India". The changes include:

- A section on authors' contributions have been added.
- The section of competing interests has been revised.

Sincerely,

Kirti Iyengar
Senior Coordinator, ARTH
On behalf of all authors of the manuscript